Author's response to reviews

Title: Surgical complications in neuromuscular scoliosis operated with posterior-only approach using pedicle screw fixation.

Authors:

Hitesh N Modi (hnm7678@yahoo.co.in)
Seung-Woo Suh (spine@korea.ac.kr)
Jae-Hyuk Yang (gurospine@naver.com)
Jae-Woo Cho (modispine@yahoo.co.in)
Jae-Young Hong (gpstylep@hanmail.net)
SuryaUdai Singh (hnm7678@yahoo.co.in)
Sudeep Jain (sudeep_im@yahoo.co.in)

Version: 8 Date: 28 April 2009

Author's response to reviews:

We have edited our accepted article as per advice by the production team. We have done all major and minor revision requirements.